Author’s response to reviews

Title: Study on Psychoeducation Enhancing Results of Adherence in patients with Schizophrenia (SPERA-S): Study protocol for a randomized controlled trial

Authors:

Donatella R Petretto (drpetretto@unica.it)
Antonio Preti (apreti@tin.it)
Carlo Zuddas (carlozuddas0@gmail.com)
Franco Veltro (franco.veltro1@virgilio.it)
Marco BL Rocchi (marco.rocchi@uniurb.it)
Davide Sisti (davide.sisti@uniurb.it)
Valentina Martinelli (valemartin@tiscali.it)
Mauro G Carta (mgcarta@tiscali.it)
Carmelo Masala (masala@unica.it)

Version: 9 Date: 28 August 2013

Author’s response to reviews:

Title: Study on Psychoeducation Enhancing Results of Adherence in patients with Schizophrenia (SPERA-S): Study protocol for a randomized controlled trial

Revision R3

Replies to the reviewers

Thank you for forwarding the reviewers’ comments and for giving us another opportunity to further improve our submission. In the sections below we have addressed each of the comments made by the reviewers. Changes to the original submission are highlighted in the text using a red font.

Referee 2:
1. P(age) 6, p(aragraph) 3, l(ine) 2. Delete [in order] in front of [to] as the words are redundant in English. Also P 18, p 3, l 2.
a. Edited as suggested.

2. P 12, p 4, l 2. Replace [ranged] by [varied].
a. Edited as suggested.

a. Edited as suggested.

4. P 25, Reference 68, l 3. Replace [[Italian]] in the middle of the l to [(Italian)] at the end of the l.
a. Edited as suggested.